Author’s response to reviews

Title: Mechanistic Study of the Cause of Decreased Blood 1,25-Dihydroxyvitamin D in Sepsis

Authors:
Chih-Huang Li (chhli2002@gmail.com)
Xiaolei Tang (xitang@llu.edu; xtang25@hotmail.com)
Samiksha Wasnik (swasnik@llu.edu)
Xiaohua Wang (Wangxiaoh5923_cn@126.com)
Jintao Zhang (jtzhang@zzu.edu.cn)
Yi Xu (Dyxu@llu.edu)
Kin-Hing Lau (William.lau@va.gov)
Bryant Nguyen (HBNguyen@llu.edu)
David Baylink (dbaylink@llu.edu)

Version: 3 Date: 26 Sep 2019

Author’s response to reviews:

Dear editors,

Thank you for allowing us to make this final revision so that this manuscript can be published in this journal. We have made modifications according to the editorial comments as shown in the following:

Editor Comments:
1. Please change the Declarations section heading “Ethics Approval” to “Ethics and consent to participate”.
 Changed.
2. If the need for consent was waived by an ethics committee or is deemed unnecessary according to national regulations, please clearly state this, including the name of the committee or a reference to the relevant legislation.

It is correctly stated.

3. In the Availability of data and materials section please clearly state who should be contacted if someone wants to request the data.

Requested contents added.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Done as instructed.